Author’s response to reviews

Title: Efficacy and safety of target combined chemotherapy in advanced gastric cancer: a Meta-analysis and system review

Authors:

kun zou (ZK907030@163.COM)
Xiong Bin (binxiong1961@whu.edu.cn)
Shuailong Yang (324317952@qq.com)
Liang Zheng (1258523913@qq.com)
Chaogang Yang (598117497@qq.com)

Version: 2 Date: 24 Aug 2016

Author’s response to reviews:

Dear Editors and Reviewers,

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Efficacy and safety of target combined chemotherapy in advanced gastric cancer: a Meta-analysis and system review”. Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our research. We have studied comments carefully and have made correction, which we hope meet with approval.

The revisions made in the revised manuscript are listed as follows:

1/ Authors' contributions: According to your comment and requests, the statement that all authors have read and approved the final manuscript was added.

We have tried our best to revise our manuscript according to the comments. These changes will not influence the content and framework of the paper.

Thank you for taking time to review our manuscript and give valuable comments. We have made correction according to the editorial requests. Special thanks to you for your good comments.

We appreciate for Editors and Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Bin Xiong